Revisions:

(1) **Abstract:** We added blank lines between the sections "Background", "Methods", "Results", and "Conclusions". We further changed the € sign in the abstract to "Euro" and separated the "Keywords" section from the abstract to make the structuring of the abstract clear. Our abstract has 218 words.

(2) **Acknowledgements:** We entered a blank section "Acknowledgements" at the end of the paper. All people who contributed to the paper are mentioned as authors, as they meet the criteria for authorship. Contributions of authors are mentioned in the "Authors’ contributions" section. The medical doctors Christian Lausmann and Rainer Wilkesmann performed the medical writing themselves and the study had no additional scientific (medical) writers to acknowledge. We added a section "funding of the study" which declares that the study had no funding.

Thank you for your comments.